Author's response to reviews

Title: Overexpression of Snail is associated with lymph node metastasis and poor prognosis in patients with gastric cancer

Authors:

Na Ri Shin (pnuhnari@hanmail.net)
Eun Hui Jeong (exile2J@naver.com)
Chang In Choi (getz0622@naver.com)
Hyun Jung Moon (moondown@pnuh.co.kr)
Chae Hwa Kwon (chkwon@pusan.ac.kr)
In Sun Chu (chu@kribb.re.kr)
Gwang Ha Kim (doc0224@chollian.net)
Tae Yong Jeon (111160@pusan.ac.kr)
Dae Hwan Kim (dh2-kim@hanmail.net)
Jae Hyuk Lee (jhlee@chonnam.chonnam.ac.kr)
Do Youn Park (pdy220@pusan.ac.kr)

Version: 3 Date: 30 October 2012

Author's response to reviews: see over
October 30, 2012

Dear editor;

We submit a revised manuscript by Shin et al. entitled "Overexpression of Snail is associated with lymph node metastasis and poor prognosis in patients with gastric cancer" in your journal. The manuscript has been revised as pointed out by the editor. The correction part in description has been highlighted. There is a point-by point description of the changes below.

comment from Editorial Board:

- The author has not fully discussed the comment made by Professor Liu regarding the effect that postoperative chemotherapy could have on the prognosis. The author should discuss this important point further in the discussion section.
  
  → We agree with editor's suggestion. We described detailed information about pre-operative and post-operative chemotherapy in the materials and methods section. Also we briefly discussed in the discussion section.

- In addition, P=0.0000 should change to P<0.0001 in manuscript
  
  → We changed all p=0.000 to p<0.001

Sincerely yours,

Do Youn Park, MD, PhD,

Department of Pathology, Pusan National University Hospital and Pusan National University School of Medicine, 1-10 Ami-Dong, Seo-Gu, Busan, Korea, 602-739
Tel: 82-51-240-7717
Fax: 82-51-256-0788
E-mail: pdy220@pusan.ac.kr